Author's response to reviews

Title: Association between Wanting Circumcision and Risky Sexual Behaviour in Zimbabwe: Evidence from the 2010-11 Zimbabwe Demographic and Health Survey

Authors:

Antony Chikutsa (antony.chikutsa@gmail.com)
Alfred C Ncube (ncubeac@gmail.com)
Shephard Mutsau (mutsaublack@gmail.com)

Version: 2 Date: 22 January 2015

Author's response to reviews: see over
Dear Editor

Re: MS: 2057003200138929 - Association between Wanting Circumcision and Risky Sexual Behaviour in Zimbabwe: Evidence from the 2010-11 Zimbabwe Demographic and Health Survey

We wish to submit a revised manuscript following reviewers’ comments. The following revisions were made in line with reviewers’ comments:

1. The abstract was revised to reflect the problem and also show the measures of association in the results section.
2. The section on data and methods was also revised to give additional details on data analysis.
3. We added a section of the conceptual framework.
4. The results section was extensively revised by adding more details on the findings and also by showing the measures of association.
5. We inserted a table which shows the n values on the four measures of risky sexual behavior.
6. We expanded the discussion section to reflect the broad findings in the analysis.

We look forward to your favourable response.

Sincerely,

[Signature]

Antony Chikutsa